

# **Earth's Future**

## **RESEARCH ARTICLE**

10.1029/2021EF002576

#### **Key Points:**

- Sea-level projections can be sorted into families, based on similarities in methods and data
- Threshold analysis shows that for certain amounts of sea-level change it is a matter of *when*, not *if*
- Combining families and thresholds leads to three categories, each with recommendations for choices to be made, to assist decision makers

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

A. B. A. Slangen, aimee.slangen@nioz.nl

### Citation:

Slangen, A. B. A., Haasnoot, M., & Winter, G. (2022). Rethinking sea-level projections using families and timing differences. *Earth's Future*, 10, e2021EF002576. https://doi. org/10.1029/2021EF002576

Received 30 NOV 2021 Accepted 28 MAR 2022

#### **Author Contributions:**

Conceptualization: A. B. A. Slangen, M. Haasnoot Formal analysis: A. B. A. Slangen, M. Haasnoot, G. Winter Investigation: A. B. A. Slangen, M. Haasnoot, G. Winter Methodology: A. B. A. Slangen, M. Haasnoot Software: A. B. A. Slangen, G. Winter Validation: A. B. A. Slangen, G. Winter Visualization: A. B. A. Slangen, M. Haasnoot, G. Winter Writing - original draft: A. B. A. Slangen, M. Haasnoot, G. Winter Writing - review & editing: A. B. A. Slangen, M. Haasnoot, G. Winter

© 2022 The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# **Rethinking Sea-Level Projections Using Families and Timing** Differences

### A. B. A. Slangen<sup>1</sup>, M. Haasnoot<sup>2,3</sup>, and G. Winter<sup>2</sup>

<sup>1</sup>Department of Estuarine and Delta Systems, NIOZ Royal Netherlands Institute for Sea Research, Yerseke, The Netherlands, <sup>2</sup>Deltares, Delft, The Netherlands, <sup>3</sup>Department of Geosciences, Utrecht University, Utrecht, The Netherlands

**Abstract** In the recent decade, many global and regional sea-level rise (SLR) projections have been published, which raises questions for users. Here, we present a series of strategies to help users to see the forest for the trees, by reducing the number of choices to be made. First, we use the similarities in the methodologies and contributing sources of the projections to group 82 projections from 29 publications into only 8 families. Second, we focus on the timing of reaching several global mean SLR thresholds, and the uncertainty therein, rather than the projected value in the year 2100. Finally, we combine the information on timing and families to define three categories which can support decision making. For global mean SLR up to 0.50 m the differences in timing are small, regardless of climate scenario or family, clearly indicating a timing window for adaptation decisions. For larger global mean SLR (0.75–1.00 m), the climate scenario becomes more important for the uncertainties in timing, but the SLR threshold will be crossed within a limited window of time, supporting adaptation decision making on the medium to long term. Beyond 1.00 m the differences between the families and climate scenario strongly determine the uncertainties in timing, and more information is needed, for instance using early warning signals, before decisions for adaptation can be made. We make recommendations on how each of these three categories, combined with the lead time and lifespan of adaptation options, can inform decisions on adaptation strategies using adaptation pathways planning.

**Plain Language Summary** Many global and regional sea-level rise projections have been published recently. As a consequence, it can be difficult to see the differences and to decide which projection to use, for instance for coastal planning. We therefore present a couple of strategies to help users see the forest for the trees in this large set of projections. First, we make a "family tree" of sea-level rise projections, based on the similarities and differences between projections. This reduces 82 projections to only 8 sea-level families. Next, we show when different sea-level thresholds (0.25 m, 0.5 m, etc.) will be crossed in each of the families. We then combine the information on the families and the timing to define three categories, where each category comes which a recommendation of the type of choices that need to be made for adaptation decision making.

### 1. Introduction

Sea-level rise (SLR) projections are required for long-term decision making in coastal zones. Many different global and regional sea-level projections for the 21st century have been published in the past years (Garner et al., 2018). Differences between sea-level projections result from (a) choices in modeling methodology (e.g., process-based, semi-empirical, reduced-complexity), (b) choices in the climate scenarios used to drive the models (e.g., Representative Concentration Pathways (RCP) (Moss et al., 2010), Shared Socio-economic Pathways (SSP) (O'Neill et al., 2017), or temperature pathways), or (c) choices in the data sources for the individual contributors to SLR (i.e., using different approaches to model thermosteric change, glaciers and ice sheets).

These choices are determined by several factors, such as the purpose of the projections; are they used for for example, information gathering or decision making on coastal protection or investments? Also, the availability of data and models or the required likelihood of the projections will affect the sea-level projections (e.g., the likely range or lower/upper/high end estimates). The wide variety in choices and combinations and new insights into SLR contributions have led to a large number of projections in the literature, with different amounts of projected SLR by 2100 and beyond. One of the largest sources of uncertainty in sea-level projections is caused by uncertainties in the magnitude, amount and timing of the Antarctic ice sheet contribution (Edwards et al., 2021; Fox-Kemper et al., 2021; Oppenheimer et al., 2019; Pattyn & Morlighem, 2020).

The large number of available projections poses difficulties for decision makers (Sriver et al., 2018) and raises questions such as: "Which projections should I use?"; "Why are these (new) projections different from others?"; "Does this difference matter?." If stakeholders select one particular projection they might run into the problem that the decisions may have to be adjusted later when new literature becomes available. To avoid the problem of having to select a particular projection, most national assessments build on an ensemble mean or consensus estimate from the Intergovernmental Panel on Climate Change (IPCC (Church et al., 2013; Fox-Kemper et al., 2021; Oppenheimer et al., 2019)), for instance in national assessments for Bangladesh (General Economics Division & Bangladesh Planning Commission, 2018) or Australia (Mcinnes et al., 2015). In some countries, the IPCC projections are tailored to the local situation (for instance in The Netherlands (KNMI'2014: Attema et al., 2014; KNMI, 2021) or Norway (Simpson et al., 2015)), or the IPCC projections are combined or extended with other projections (e.g., Finland (Johansson et al., 2014) or New Zealand (Lawrence et al., 2018)). As the IPCC projections are typically updated only every five to seven years, in some cases the IPCC AR5 projections have been adjusted for new information, in particular on accelerated ice sheet contributions, for instance in the UK (UKCP18, Palmer et al., 2018) or the US (New York City Panel on Climate Change, R. Horton et al., 2015). Alternatively, in some cases scenarios are based on (a set of) single planning values, for instance in the US (Sweet et al., 2017) or France (Le Cozannet et al., 2017).

The IPCC regularly provides an assessment of the latest research on sea-level change, and combines the existing methodologies and data into an assessed set of sea-level rise projections (Church et al., 2013; Fox-Kemper et al., 2021; Oppenheimer et al., 2019). Before IPCC AR6, these projections typically were central range projections, resulting from a combination of the necessity for scientific consensus and a lack of literature for projections outside the *likely* range. The lastest IPCC report, AR6 (Fox-Kemper et al., 2021), for the first time also provided projections for a low-probability, high-impact scenario, which includes *low confidence* processes of accelerated ice sheet mass loss in Greenland and Antarctica. As a result, the latest IPCC projections provide a larger range of possible futures and associated uncertainties in sea-level change.

In this manuscript, we discuss what the uncertainties in sea-level projections mean for different planning time horizons (mid-century, end-of-century and beyond). We also present a set of strategies which can provide guidance on how to use and interpret the available SLR projections. Our first step is to structure the large number of SLR projections by using the genealogy of SLR projections to define families by grouping projections that have a number of methods and/or SLR contribution approaches in common (Section 2), inspired by the work of Knutti et al., (2013) and Masson and Knutti (2011). Our second step is to rethink the information from the projections by reversing the information on the uncertainty and asking *when* a certain sea-level threshold will be crossed, in addition to *which* or *if* a certain sea level will be reached in 2100 (Section 3). Finally, we combine the information on the families and their timing into three categories, each of which require different types of decisions by adaptation decision makers (Section 4).

The notion that adaptation to SLR is a matter of *when* to adapt and not *if* has, for example, already been acknowledged for the U.S. Atlantic and Gulf coast (Hauer, 2017) where relocation may seem inevitable (Keeler et al., 2018). The importance of uncertainty in the timing of reaching different sea levels for coastal adaptation planning was acknowledged in the recent IPCC AR6 Working Group 1 report (Arias et al., 2021). Decision making and implementation of such drastic measures or large coastal defence structures requires a long lead time (SLR Box in Cooley et al., 2022; Haasnoot et al., 2020) and highlights the need to inform decision-makers early on about the projected timing of crossing a certain SLR threshold rather than a projected SLR magnitude at an arbitrary time.

The overall aim of this work is to help decision makers 'see the forest for the trees' by providing insight into uncertainties on different time horizons and give tools on how to best deal with them in planning. The recommended starting point for decision makers is Section 4, where three categories of SLR thresholds are presented, with recommendations for different types of decisions that need to be taken in each category (Section 4.1). We then provide a flow chart with questions to assist decision makers in dealing with different bandwidths of the timing of SLR in relation to different types of lead times (Section 4.2).

23284277, 2022, 4, Downle



### 2. Data and Methods

The first step to restructure the large number of SLR projections (Section 2) is inspired by work done on the genealogy of climate models in the Climate Model Intercomparison Project (CMIP) database (Knutti et al., 2013; Masson & Knutti, 2011). Their aim was to determine how (in)dependent climate models in an ensemble are, as the models sometimes have components in common, for instance for the oceanic or atmospheric parts. Here, our aim is to use the genealogy of SLR projections to define families, by grouping projections that have a number of methods and/or SLR contribution approaches in common.

### 2.1. Selecting the Sea-Level Projections

All sea-level projections used here are from publications from 2013 onwards (when IPCC AR5 (Church et al., 2013) was published), use CMIP5 data and are based either on RCP scenarios or on temperature pathways. The only exception is most recent projection from IPCC AR6 which used SSP-RCP scenarios and CMIP6 information. We only use global mean sea-level projections from publications that provide information on the sources and methods of the individual contributions. This leads to a total of 82 projections from 29 publications: Bakker et al., 2017; Bamber et al., 2019; Le Bars et al., 2017; Bittermann et al., 2017; Buchanan et al., 2017; Carson et al., 2016; Church et al., 2013; Fox-Kemper et al., 2021; Goodwin et al., 2017; Grinsted et al., 2015; Jackson et al., 2016; Nauels, Meinshausen, et al., 2017; Nauels, Rogelj, et al., 2017; Oppenheimer et al., 2019; Palmer et al., 2020; Perrette et al., 2013; Rasmussen et al., 2018; Schleussner et al., 2016; Slangen et al., 2014; De Winter et al., 2017; Wong et al., 2017.

The projections are distributed over five climate scenarios of RCP and global warming levels: RCP2.6 (16); RCP4.5 (19); RCP8.5 (29); 1.5° (7); 2° (8). For the analysis, the 1.5° scenario is combined with RCP2.6 and SSP1-2.6, the 2.0° scenario with RCP4.5 and SSP2-4.5, and RCP8.5 with SSP5-8.5 (Collins et al., 2013; Fox-Kemper et al., 2021).

In some publications, multiple projections are presented with different methods/sources: Bakker et al., 2017 (2), De Winter et al., 2017 (2), Fox-Kemper et al., 2021 (2), Goodwin et al., 2017 (3), Jackson & Jevrejeva, 2016 (2), Jevrejeva et al., 2018 (2), Jackson et al., 2018 (2) and Le Bars et al., 2017 (3). On the other hand, the projections presented in Slangen et al., 2014 and Carson et al., 2016 are the same, and the projections in Kopp et al., 2014 and Buchanan et al., 2017 are the same; each projection is therefore only included once.

For the comparison of all projections in 2100 (Section 3.1), we primarily use SLR projections from Garner et al., 2018 (their Table S1 in Supporting Information S1), who applied a linear approach to compute the SLR over 100 years (SLR<sub>100</sub>) for all projections [SLR<sub>100</sub> = SLR<sub>pub</sub>\*(100/( $y_{end}-y_{beg}$ ))], using the SLR from the original publication (SLR<sub>pub</sub>) and the published begin year ( $y_{beg}$ ) and end year ( $y_{end}$ ). For more recent publications, the SLR is taken from the original publications following the same approach to compute the change over a 100 year period (Data Set S1). Missing 5th and 95th percentiles were calculated using the standard deviation of the dataset and assuming a normal distribution. This assumption may lead to an underestimation of the 95th percentile for the two publications which only provided 17th and 83rd percentiles and are non-Gaussian: SROCC, 2019 (Oppenheimer et al., 2019) (in Sections 3.1 and 3.2) and Perrette et al., 2013 (in Section 3.1).

For the SLR threshold analysis which assesses the timing at which a particular SLR threshold is exceeded (Section 3.2), we use a subset of publications which provide time series rather than only the difference over the 21st century (Data Set S2). We use one time series to represent a family if all family members are alike, or multiple time series to encompass all family members if required. Only five of these time series extend beyond 2100 and are used for threshold analysis up to the year 2200.

### 2.2. Sorting the Sea-Level Projections Into Families

For all available publications, we first analyze the treatment of four major contributors to SLR: the Greenland ice sheet, the Antarctic ice sheet, glaciers and ocean warming (thermosteric change). Each of these are modeled using a variety of methods across the literature, as explained in Text S1 and Figure S1 in Supporting Information S1. There is less variety in the treatment of the terrestrial water storage change contribution in the available



**Figure 1.** Decision tree for sea-level families. Blue boxes indicate main questions (Q1 and Q2), green boxes indicate follow-up questions, white boxes indicate families. To show that this is not meant as a static framework, two example boxes for new families are indicated by the dashed arrows, but new families can be added in other places as well. See Text S1 and Figure S1 in Supporting Information S1 for additional details.

projections, and this is therefore not taken as a distinguishing factor for the sorting. Using the information on the four main contributors, we set up a decision tree (Figure 1), which allows us to sort the projections into families. Based on the analysis of the methods and data used in the projections, we identify eight different families, as detailed below.

The first question in the decision tree relates to the treatment of the ice sheet contributions. We identify two families relying on structured expert judgement (SEJ) studies: family 5 (SEJ) which uses the SEJ results directly and family 6 (shifted SEJ) which uses the SEJ to inform the shape of the ice sheet probability distribution. All publications which use the Antarctic estimates of Deconto and Pollard (2016) are in family 8 (DCP16).

The second question in the decision tree relates to the modeling approach: process-based or semi-empirical. We identify two IPCC-related families relying primarily on process-based modeling, where the difference is that family 1 uses a combination of multiple ice sheet and glacier estimates (direct IPCC), while family 2 uses individual glacier or ice sheet estimates (blended IPCC). There are three different families using semi-empirical (SEM) type approaches: family 3 (BRICK), family 4 (SEM), and family 7 (MAGICC/SEM) which uses SEM methodology in combination with the reduced-complexity model MAGICC (Meinshausen et al., 2011).

The families can be used for different purposes. For instance, families 1 and 2 are primarily based on IPCC and therefore represent a consensus-based estimate. Families 5, 6 and 8 all feature skewed probability distributions for the Antarctic ice sheet contribution, and may be used by decision makers that are particularly interested in low-probability/high-risk SLR. If one would like to use the outcomes of fully coupled climate models primarily, families 1 and 2 might be most suitable, whereas if one would like to be able to use and reproduce the SLR projections, reduced-complexity models in families 3, 4 and 7 might be the best choice. Families 5 and 6 may inform uncertainties in SLR as a result of processes potentially still missing from models.

4 of 16

The decision tree presented here reflects current knowledge, but we stress that it is not meant to be a static framework: new families may be added when new data, methods or models become available (indicated by a few example "new" boxes in Figure 1), while other families may become obsolete when there are no more new developments in a family. An example of a start of a potential new family are the IPCC AR6 projections, which are currently added to family 1. These are based on CMIP6 models rather than CMIP5 models, and when more publications with CMIP6 data become available they might branch off into their own family.

# 3. Rethinking the Sea-Level Projections

With the sea-level projections sorted into families, the second step in our approach is to rethink the information that can be derived from the projection families. In addition to knowing *which* sea level will be reached in 2100, as is often the focus in SLR publications, it might be helpful to reverse the information about the uncertainty, and assess *when* a certain sea level threshold will be exceeded. The timing of a particular magnitude of SLR in the SLR projection families may demonstrate that the question is not *if* to adapt but *when* to adapt. Thus, in addition to presenting uncertainty as a range in the magnitude of SLR at a certain point in time (*which/if*, Section 3.1), we also present uncertainty as a range in the timing (a period) for crossing a particular SLR threshold (*when*, Section 3.2).

### 3.1. Sea-Level Family Projections for the Year 2100

The projected SLR in the year 2100 (Figure 2) shows an overall good agreement within families for all three scenarios. For the RCP2.6 scenario (Figure 2a) the differences between the families are the smallest of all three scenarios. There is only one projection outside the overall 5%–95% probability range (the black dashed lines): in family 7 (MAGICC/SEM) Perrette et al., (2013) presents a large SLR projection, due to relatively large contributions from the Greenland and Antarctic ice sheets (0.22 and 0.23 m, respectively), which is about three times larger than the ice sheet contributions in the other family 7 projections. The reason for this is that in Perrette et al. (2013) the ice sheets are taken as the residual of the total sea-level change computed with a semi-empirical model, minus the steric and glacier contribution. In the other family 7 projections the ice sheet contribution is explicitly parameterized, leading to smaller ice sheet contributions.

The differences between and within the families increase for the RCP4.5 scenario, but the overall agreement is still high (Figure 2b). Families 3 (BRICK) and 8 (DCP16) project a larger amount of SLR compared to the other families, illustrating the uncertainty in SLR projections as a result of the Antarctic ice sheet contribution. Within family 8 (DCP16), Kopp et al. (2017) directly used the DCP16 estimates leading to the higher SLR projection, whereas Nauels, Rogelj, et al. (2017) used the DCP16 estimate to calibrate their SEM and arrived at a slightly lower SLR projection.

For the RCP8.5 scenario, the differences between and within the families are the largest, and mainly families 3 (BRICK) and 8 (DCP16) stand out (Figure 2c). For family 8, the projected Antarctic contribution used from DCP16 is either 0.64 m  $\pm$  0.49 (Bakker et al., 2017; Kopp et al., 2017; Nauels, Rogelj, et al., 2017) or 1.05  $\pm$  0.30 m in 2100 (Le Bars et al., 2017), due to differences in the ice model configuration, which leads to SLR projections that are generally larger than the SLR projections in the other families.

When considering the families over all three scenarios, the medians of families 4 (SEM), 5 (SEJ), 6 (SEJ-shifted) and 7 (MAGICC/SEM) are close to the ensemble-mean for all scenarios. Families 3 (BRICK) and 8 (DCP16) are always above the overall ensemble-mean, in particular for RCP8.5, and the upper 95% probability is larger due to a skewed distribution of higher projected contributions from the Antarctic ice sheet.

Families 1 (Direct IPCC) and 2 (Blended IPCC) are generally below the median of all the projections, which is in line with previous analyses that the assessment methodology of IPCC tends to underestimate rather than overestimate the projected changes in the climate system (Brysse et al., 2013; Garner et al., 2018; Horton et al., 2020). The medians of the IPCC AR6 projections (Fox-Kemper et al., 2021) are close to previous IPCC estimates for all scenarios, although they are based on SSP scenarios and CMIP6 models rather than RCP scenarios and CMIP5 models (see also Hermans et al., 2021). The 95th percentile of the IPCC AR6 *low confidence* projections is an outlier in family 1, which can be explained by the fact that these are based on a combination of SEJ and DCP,



# **Earth's Future**



**Figure 2.** Global mean sea-level projections for 2000–2100 (m) for eight projection families and three scenarios: (a) RCP2.6/ $1.5^{\circ}$ ; (b) RCP4.5/ $2.0^{\circ}$ ; (c) RCP8.5. Circles indicate the medians, black error bars the 5%–95% range of each projection. Filled circles are RCP scenarios, open circles are temperature scenarios, black open circles are the IPCC AR6 SSP-RCP projections. The median of each family is indicated by the colored full line, the medians of 5%–95% probabilities per family by colored dashed lines. The median of all projections for each scenario is indicated by the horizontal black full line and the median of all 5%–95% probabilities by horizontal black dashed lines. Crosses (X) indicate the global mean projection time series used for threshold analysis.

and therefore are also related to families 5, 6 and 8, with a skewed distribution to higher contributions from Antarctica.

### 3.2. Timing in Reaching Thresholds in the Sea-Level Families

We use global mean SLR projection time series to compute the crossing of different SLR thresholds (Figure 3), with at least one time series representing each family (see Section 2.1). We find that the time when a SLR threshold of 0.25 m (with respect to the year 2000) is first reached differs little across the eight families and the three RCP scenarios. All eight families project the exceedance of 0.25 m with a 50% probability for the middle of the century ( $\sim$ 2040–2060). With a 5% probability, 0.25 m is already reached before 2040 in more than half of the families, regardless of the scenario.

A SLR of 0.50 m is projected for the second half of the 21st century in all families for all scenarios (Figures 3a-3c), with at least a 5% probability across scenarios, and a 50% probability for all families in RCP4.5 (2066–2099) and RCP8.5 (2059–2081). For RCP2.6, the median exceedance of 0.50 m occurs late in the 21st or early in the 22nd century (Figure 3d). With a 5% probability, 0.50 m is exceeded before 2060 in four families (RCP2.6), in





**Figure 3.** Timing in reaching global mean sea-level rise threshold values (m). The first year in which a projected total sea-level threshold (m) is reached with respect to the year 2000, for three RCP/temperature scenarios, and projection time series representing the eight families (colors as in Figure 2), (a–c) between 2000 and 2100, (d–f) between 2000 and 2200. Vertical bars represent the median, dark colors the 17%–83% range, light colors the 5%–95% range.

six families (RCP4.5) and in seven families (RCP8.5, the eighth by 2065), and can be crossed as early as 2041 for RCP8.5. This means that, regardless of the scenario, 0.50 m SLR will occur in the second half of this century according to most projections, with RCP2.6 and RCP4.5 mainly in the fourth quarter and RCP8.5 moving up to the third quarter of the century.

For SLR thresholds beyond 0.50 m there is an increasing dependency on the climate scenario. At the end of the 21st century, a SLR of 0.75 m (with at least 5% probability) is projected for RCP2.6 in three families, for RCP4.5 in seven families and in all families for RCP8.5. A SLR of 1.00 m is exceeded before 2100 in the median projections of RCP8.5 for families 3 (BRICK), 5 (SEJ) and 8 (DCP16), and in addition in the first half of the next century (Figures 3d and 3e) in families 1 (Direct IPCC), 6 (SEJ-shifted) and 7 (MAGICC/SEM). For RCP4.5, families 1, 6, and 7 reach 1.00 m in the second half of the 22nd century (50% probability), but 1.00 m can already occur as early as 2069 in family 8 (RCP8.5, 5% probability).

For thresholds over 1.00 m SLR the projections between families show increasing divergence, in particular for the RCP8.5 scenario (Figures 3c and 3f), where families 3 (BRICK), 5 (SEJ) and 8 (DCP16) tend to reach higher thresholds earlier than other families. For instance, SLR reaches 1.50 m around 2100 in family 8 for RCP8.5 (median), and this does not happen until several decades later in families 1, 6 and 7 (Figure 3f). This divergence between families continues for SLR thresholds above 2.00 m.

Overall, the likely (17%–83%) and very likely (5%–95%) time windows for SLR to exceed any of the defined thresholds are widest for RCP2.6 (Figures 3a and 3d) and narrowest for RCP8.5 (Figures 3c and 3f), as sea level rises faster for warmer climate scenarios. The narrowest time range for each threshold is projected in family 3 (BRICK) and the widest in family 2 (Blended IPCC). Up to 0.5 m of SLR, the timing in threshold exceedance is within a range of 20–30 years for the median values of the projections in all eight families.

For some purposes it may be more relevant to know the timing of reaching a certain SLR rate rather than a cumulative threshold, for instance when determining the feasibility of soft and nature-based adaptation measures. For example, beach nourishments are temporary adaptation measures and their lifespan can be compromised by higher SLR rates, which require larger or more frequent nourishments (Haasnoot et al., 2020). Coastal vegetation can trap and stabilize sediments (Nepf, 2012), but vegetation needs time to grow and keep up with SLR (Kirwan et al., 2016), which may not be possible for higher SLR rates. To illustrate the timing differences in SLR rates, Figures S2–S4 in Supporting Information S1 provide the timing of different SLR rates (mm/yr) for the 5th, 50th and 95th percentiles.

# 4. Connecting SLR Families and Timing Differences to Adaptation Decision Making

We now combine the information on SLR timing and uncertainty in the different families to define categories of SLR thresholds which each require different types of decisions (Section 4.1). We also discuss how the global framework presented in this study can be a starting point for local applications (Section 4.1). Finally, we provide a flow chart with questions to assist decision makers in dealing with the uncertainties in timing of SLR (Section 4.2).

### 4.1. Defining Categories of Sea-Level Thresholds

Based on the information from the analysis of the families and timing differences, we define three categories of SLR thresholds. Each category provides a recommendation for the type of choices required when it comes to selecting a sea-level projection family and scenario:

*Category 1*: a small bandwidth in timing, for low SLR thresholds (up to 0.50 m). There is little uncertainty that this threshold will be crossed within the foreseeable future, irrespective of the projection family chosen or the climate scenario followed. This means that, for decisions on adaptation measures to 0.50 m, decision makers do not need to worry about which SLR family or even which individual projection they choose, they can use whichever is accessible to them.

*Category* 2: a moderate bandwidth in timing, for SLR thresholds between 0.50 and 1.00 m. Here, the uncertainty in timing mainly results from different climate scenarios. In this case it is important to realise that if the timing of crossing the threshold is within the envisioned lifespan of a measure or the time horizon of the decision analysis, it is a matter of *when* and not *if* this SLR threshold will be crossed. For adaptation measures within this bandwidth, this means that the main uncertainty is in the choice of the climate scenario, and less in the choice of the SLR projection family.

*Category 3*: a large bandwidth in timing, for SLR thresholds beyond 1.00 m, with uncertainty in timing both resulting from different climate scenarios and from projection families. In this situation, gathering more information on the likelihood of each of the climate scenarios could reduce the uncertainty in the timing. If the timing is beyond the lifespan of a measure or the time horizon of the decision analysis, this SLR amount is likely not relevant for this particular measure. For decision makers, it now becomes relevant to choose a particular family and climate scenarios, for instance based on their risk aversion, as the choice impacts the timing in crossing SLR thresholds. However, as the timing is further into the future, there may still be time to adjust the adaptation decision when more information about climate scenarios and uncertainties in SLR contributions becomes available.

We use a conceptual example to show how the timing bandwidths of SLR thresholds (Figures 4a and 4b) can be used to assess when signals may be detected that can inform adaptation and for how long present and future investments and measures can perform acceptably (i.e., contain risk) as sea levels rise. The latter is sometimes referred to as an adaptation threshold or tipping point (Barnett et al., 2014; Haasnoot et al., 2013; Kwadijk et al., 2010). The period over which such an adaptation tipping point is reached can be considered as the functional lifespan of a measure. Beyond an adaptation tipping point, additional or new measures may be needed. Alternative sequences of measures, or adaptation pathways, can be visualized into a route map or adaptation map (Haasnoot et al., 2013) (Figure 4c). The lead time needed for planning and implementation of follow-up measures can inform decision makers about when decisions need to be made at the latest (Figure 4d). By comparing the timing of SLR threshold exceedance (in families or individual projections) to the lead time and functional lifespan of adaptation options, decision makers can assess whether it is justified to make the adaptation investment.

Although the analyses here are based on global mean SLR projections, it is important to note that adaptation decision making often relies on local information. For local applications, the global time series in Figure 4a should therefore be replaced by local sea-level projections, which can then be combined with local information on adaptation options to inform a local adaptation pathway analysis. As an example, Figure S5 in Supporting Information S1 shows the local timing of SLR threshold exceedance at four locations compared to the global timing in Figure 3. This shows that indeed there are regional differences compared to the global mean, for instance at the east coast of the US, where an above-average projected SLR leads to earlier crossing of thresholds (compared to the global mean) by about 5–10 years. However, it also shows that the global analysis gives a first order assessment of the timing of SLR threshold exceedance and that the three categories defined above can serve as a starting point for local decisions. We do note that not all of the projections included in the current analysis provide sufficient regional or local information that would allow a similar analysis for all families. For instance, semi-enclosed basins are often poorly represented in climate models and would therefore require an additional downscaling step to allow for a local analysis (e.g., Hermans et al., 2020). There are also publications which only provide global mean projections, for instance many of the semi-empirical projections in Family 3, which would require a regionalization step in order to be used in regional analyses.

### 4.2. Using SLR Families and Timing to Inform Adaptation Decision Making

When it comes to adaptation decision making, the timing of SLR thresholds in the different families can be compared against the potential functional lifespan of present and future measures under multiple futures. To assist with the types of questions to ask and decisions to take, we present a flowchart (Figure 5). We identify three main uses of the sea-level families and their timing in threshold exceedance.

The first use of our analysis is to assess whether or not it is a matter of time (*when* instead of *if*). The timing bandwidth of threshold exceedance corresponds with the shortest to longest functional lifespan and thus also indicates if the uncertainty about the functional lifetime is (relatively) small or large (first question in Figure 5). For example, a coastal defense structure designed for 0.50 m of SLR could be sufficient for the next 40–75 years



# **B** Range SLR threshold



# C Adaptation Pathways



# D Lead time and early warning signals





23284277, 2022, 4. Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2021EF002576 by Utrecht University Library. Wiley Online Library on [12/10/2023], See the Terms and Conditions

(https://onlinelibrary.

conditions) on Wiley Online Library for rules of use; OA articles

are governed by the applicable Crea

Licens

(median values), but might already reach the end of its functional lifetime within 25 years (2045) under RCP8.5 in some families. Comparing this range in timing with the envisioned lifespan helps to evaluate the consequences and whether or not to invest in a particular measure. For instance, a small uncertainty in timing (middle blue box, Figure 5) that occurs within the envisioned lifespan of a measure will reduce the functional lifespan (right blue box, Figure 5). With a moderate to large uncertainty in timing that is inside the envisioned lifespan of a measure, it is a matter of time. If the threshold exceedance occurs partly outside the envisioned lifespan of an investment (middle light green box, Figure 5), more information about for instance climate mitigation and the relevance of particular climate scenarios (e.g., through learning over time) could reduce uncertainty. If this is only occurring for some climate scenarios, climate mitigation could, to some extent, still avoid lifespan reduction. The above is not only relevant for a particular measure, but also for an adaptation plan and its lifespan and thus the time horizon of a decision analysis.

The second use of our analysis lies in assessing which measures to take. Building upon the notion that the functional lifespan of an investment is determined by exceeding a certain amount or rate of SLR, the timing in SLR families can help to assess which adaptation measure would be needed given the preferred lifetime of an investment or the preferred frequency of adaptation. Large coastal defense structures such as the Dutch Delta works have a long lifespan (typically 50–200 years (Haasnoot et al., 2020; Hallegatte, 2009)), but also other adaptation strategies, such as floodproofing of buildings, are intended to be effective for several decades or more. The associated SLR thresholds tend to fall into category 3, which means there is a stronger dependency on the choice of SLR family and climate scenario. If the functional lifespan is considered too short because of the required economic or societal investments or impacts, this could indicate that it would be necessary to adapt to higher amounts of SLR, which would be increasingly needed under accelerating SLR (Haasnoot et al., 2020). Not only investments for coastal adaptation, but also the functioning of other investments, such as buildings or power plants, may become unacceptably affected as sea levels rise. The timing of SLR could indicate whether and when this could happen and thus affect the investments' lifespan.

In addition the functional lifespan of measures, one needs to account for the lead time of potential follow-up measures to assess whether and when to start preparing for them. Decision makers with a very low risk tolerance (Hinkel et al., 2019; Oppenheimer et al., 2019) may want to focus on the earliest moment of threshold exceedance in SLR family 8 (red boxes in Figure 5). The lead time needed for planning and implementation of coastal defense measures can take years to decades (Haasnoot et al., 2020) and time may also be needed for research, experiments and technological developments, especially when hard limits are exceeded and step-changes are needed. In practice, the preparation time will also depend on the political, societal and institutional context. Overlap in bandwidth of the timing of relevant thresholds for adaptation could complicate adaptation, as preparation for a measure may need to start when its predecessor is still being implemented. This becomes an increasingly relevant problem under accelerating SLR (Haasnoot et al., 2020).

The third use of our analysis is to indicate whether the timing of SLR thresholds in the different families can be used to evaluate SLR indicators on their potential to serve as early warning signals and inform decision-making (Haasnoot et al., 2018; Stephens et al., 2018). Ideally, signals on changing conditions would leave enough time for planning and implementation of coastal defense measures and would thus be detected reliably before a decision point (Haasnoot et al., 2018). For example, in Figure 4 the projected timing of a threshold exceedance of 0.25 m might not provide enough time to implement a follow-up measure of the yellow measure under RCP8.5. A risk-averse decision-maker could decide to adjust the plan (e.g., replace the measure protecting against 0.50 m by a measure protecting against 0.75 m) or to make preparations to reduce the lead time (e.g., regulations or spatial planning). The concept of *learning scenarios*, as introduced by (Hinkel et al., 2019) could be used to describe the effect of the timing of new information becoming available on reducing the time range for the threshold

**Figure 4.** Conceptual example of threshold exceedance and decision making. (a) Example sea-level rise projections (blue and red, median and 5%–95%), (b) the timing of reaching 0.25, 0.50 and 1.50 m SLR thresholds in the projections in panel (a). Time series can be replaced by a global or regional SLR time series from any family, depending on the required application. Threshold timing determines the functional lifespan of a measure in (c) the adaptation pathways map (the length of the colored line). The measure designed for 0.25 m SLR (category 1, green) reaches an adaptation tipping point ~2045–2055 (based on the median values) and could be followed by a measure for 0.50 m of SLR (category 2, yellow) which reaches an adaptation tipping point ~2070–2090. Alternatively, a switch to a measure for 1.50 m of SLR is possible (category 3, dark blue), which only reaches an adaptation tipping point in RCP8.5 in the next 130 years. (d) The adaptation tipping points (circles) combined with the lead times indicate when a decision for a measure associated with the yellow and blue pathways needs to be taken (triangles) in order to have enough time for planning and implementation of a measure and contain the risk of SLR (lead time can be adjusted for specific measures). Confronting the timing of the SLR thresholds from (b) with the lead time helps to assess if a particular threshold exceedance can be used as an early warning signal for adaptation.



# How can information on the uncertainty in timing of SLR help decision making?



Figure 5. Flow chart illustrating how the bandwidth in timing of threshold exceedance can be used for decision making while considering the envisioned lifespan of investments, the lead time for planning and implementation as well as the risk tolerance of decision makers.

exceedance. Particularly for categories 2 or 3, the observed SLR may also indicate which of the projection families is being followed, when climate scenarios start to diverge, or if an adaptation tipping point is approaching.

### 5. Conclusions

In this paper, we have proposed a series of strategies to rethink the growing set of sea-level projections and help decision makers "see the forest for the trees". First, we restructured the large number of projections by grouping 82 projections of 29 publications into 8 families, based on similarities in the choice of methodology and contributing sources. Second, we analyzed the timing of reaching certain SLR threshold levels, and found that there is a remarkable agreement in the timing of when a certain threshold will be first reached in each of the temperature and RCP scenarios.

Finally, we combined the analysis of SLR families and their threshold exceedance times to define three categories, where each category requires different choices to be made by decision makers. Category 1 includes the smaller SLR thresholds (up to 0.50 m), where the small differences in timing show that adaptation to these levels will have to happen in the foreseeable future anyway, regardless of the projection family used or the climate scenario realised. Category 2 includes the moderate SLR thresholds (0.75–1.00 m), where the climate scenario becomes more important, but the divergence between the families is still limited. SLR thresholds beyond 1.00 m fall into category 3, where the differences between the families, in combination with the climate scenario realised, critically determine the uncertainties in timing of reaching a certain threshold. For decisions with a longer lifespan or for long-term planning, SLR in category 3 may still be within their lifespan and it can therefore still be a matter of *when* and not *if* adaptation is needed. For each SLR threshold, a decision maker may choose and compare different SLR families with different characteristics, based on their level of risk-aversion, on their measures' desired lifespan or on the lead time required, but we have shown that this choice becomes more relevant for the higher SLR thresholds. The timing of threshold exceedance can also be used to assess for how much SLR to adapt or invest (and thus also how to adapt), depending on the envisioned lifetime of an investment, and to assess if a SLR threshold exceedance can serve as a early warning signal to inform adaptation decision making.

The families can aid decision makers in identifying the type of sea-level projections that best match their risk tolerance level. Whereas some decision makers may prefer consensus estimates from the IPCC family, decision makers with a very low risk tolerance may want to use families that include significantly larger contributions from the Antarctic ice sheet, such as family 8. In addition, we have shown how the timing of SLR threshold exceedance in the SLR families can be used as a basis for the dynamic adaptive policy pathways approach (Haasnoot et al., 2013), which feeds into an adaptive plan consisting of short-term actions and long-term options that can be implemented as sea levels rise.

We do note that there are many more elements to sea-level change that play a role on various spatial and temporal scales (e.g., Slangen, Meyssignac, et al., 2017) and are important for local decision-making and adaptation to SLR. For one, sea-level change is not uniform across the world: large spatial differences exist as a result of differences in ocean density, mass-induced gravitational effects and vertical land movement (glacial isostatic adjustment, tectonics, subsidence) (Church et al., 2013; Fox-Kemper et al., 2021; Slangen et al., 2014, 2012; Slangen, Adloff, et al., 2017). Second, sea-level change varies on different timescales. Uncertainties related to interannual to decadal variability, for instance the El Nino Southern Oscillation, the Pacific Decadal Oscillation or the North Atlantic Oscillation (e.g., Roberts et al., 2016), are partially accounted for by using multi-model ensembles for the sea-level projections. While these responses tend to be dampened in the global mean, they will have a larger impact on sea-level change on local scales. On shorter time scales, sea-level changes locally from season to season (Hermans et al., 2019; Wahl et al., 2017)). When considering adaptation to SLR, the knowledge on local (changing) extremes therefore needs to be taken into account, by using the framework presented here in combination with more tailored information on changing local sea-level extremes or existing coastal protection measures to complete the analysis.

We recognize that the strategies presented still require the user to take some decisions, which is inevitable as science keeps on progressing and the future is never completely certain. However, we are convinced that this framework, which rethinks the sea-level projections by using families and threshold timing, can serve as an easier starting point for SLR projection users across the board.



### **Data Availability Statement**

The data used in this study (original time series and processed data for figures) are available as Data Sets S1 and S2, and on Zenodo under doi:10.5281/zenodo.6340076.

### References

- Arias, P. A., Bellouin, N., Coppola, E., Jones, R. G., Krinner, G., Marotzke, J., et al. (2021). Technical summary. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intercovernmental panel on climate change. Cambridge University Press.
- Attema, J., Bakker, A., Beersma, J., Bessembinder, J., Boers, R., Brandsma, T., et al. (2014). KNMI'14: Climate change scenarios for the 21st century-A Netherlands perspective. De Bilt.
- Bakker, A. M. R., Wong, T. E., Ruckert, K. L., & Keller, K. (2017). Sea-Level projections representing the deeply uncertain contribution of the West Antarctic ice sheet. Scientific Reports, 7(1), 3880. https://doi.org/10.1038/s41598-017-04134-5
- Bamber, J. L., Oppenheimer, M., Kopp, R. E., Aspinall, W. P., & Cooke, R. M. (2019). Ice sheet contributions to future sea-level rise from structured expert judgment. *Proceedings of the National Academy of Sciences of the United States of America*, 116(23), 11195–11200. https://doi. org/10.1073/pnas.1817205116
- Barnett, J., Graham, S., Mortreux, C., Fincher, R., Waters, E., & Hurlimann, A. (2014). A local coastal adaptation pathway. *Nature Climate Change*, 4(12), 1103–1108. https://doi.org/10.1038/nclimate2383
- Bittermann, K., Rahmstorf, S., Kopp, R. E., & Kemp, A. C. (2017). Global mean sea-level rise in a world agreed upon in Paris. Environmental Research Letters, 12(12), 124010. https://doi.org/10.1088/1748-9326/aa9def
- Brysse, K., Oreskes, N., O'Reilly, J., & Oppenheimer, M. (2013). Climate change prediction: Erring on the side of least drama? *Global Environmental Change*, 23(1), 327–337. https://doi.org/10.1016/j.gloenvcha.2012.10.008
- Buchanan, M. K., Oppenheimer, M., & Kopp, R. E. (2017). Amplification of flood frequencies with local sea level rise and emerging flood regimes. Environmental Research Letters, 12(6), 064009. https://doi.org/10.1088/1748-9326/aa6cb3
- Carson, M., Koehl, A., Stammer, D., Slangen, A. B. A., Katsman, C. A., van de Wal, R. S. W., et al. (2016). Coastal sea level changes, observed and projected during the 20th and 21st century. *Climatic Change*. https://doi.org/10.1007/s10584-015-1520-1
- Church, J. A., Clark, P. U., Cazenave, A., Gregory, J. M., Jevrejeva, S., Levermann, A., et al. (2013). Sea Level change. In T. F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), *Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change.* Cambridge University Press.
- Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., Friedlingstein, P., et al. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In T. F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), Climate change 2013: The physical science basis. Contribution of working group I to the Fifth assessment report of the intergovernmental panel on climate change. Cambridge University Press.
- Cooley, S., Schoeman, D., Bopp, L., Boyd, P., Donner, S., Ito, S.-I., et al. (2022). Chapter 3: Oceans and coastal ecosystems and their services. In *Climate change 2022: Impacts, adaptation and vulnerability.*
- Deconto, R. M., & Pollard, D. (2016). Contribution of Antarctica to past and future sea-level rise. *Nature*, 531(7596), 591–597. https://doi.org/10.1038/nature17145
- De Winter, R., Reerink, T. J., Slangen, A. B. A., De Vries, H., Edwards, T., & Van De Wal, R. S. W. (2017). Impact of asymmetric uncertainties in ice sheet dynamics on regional sea level projections. *Natural Hazards and Earth System Sciences Discussions*. 17(April), 1–25. https://doi. org/10.5194/nhess-2017-86
- Edwards, T. L., Nowicki, S., Marzeion, B., Hock, R., Goelzer, H., Seroussi, H., et al. (2021). Projected land ice contributions to twenty-first-century sea level rise. *Nature*, 593(7857), 74–82. https://doi.org/10.1038/s41586-021-03302-y
- Fox-Kemper, B., Hewitt, H., Xiao, C., & Al, E. (2021). Ocean, cryosphere, and sea-level change. In Climate change 2021: The physical science basis.
- Garner, A. J., Weiss, J. L., Parris, A., Kopp, R. E., Horton, R. M., Overpeck, J. T., et al. (2018). Evolution of 21 st century sea-Level rise projections. *Earth's Future*, 6(11), 1603–1615. https://doi.org/10.1029/2018EF000991

General Economics Division and Bangladesh Planning Commission. (2018). Bangladesh Delta plan 2100 (Bangladesh in the 21st century).

- Goodwin, P., Haigh, I. D., Rohling, E. J., & Slangen, A. (2017). A new approach to projecting 21st century sea-level changes and extremes. *Earth's Future*, 5(2). https://doi.org/10.1002/2016EF000508
- Grinsted, A., Jevrejeva, S., Riva, R. E. M., & Dahl-Jensen, D. (2015). Sea level rise projections for northern Europe under RCP8.5. *Climate Research*, 64, 15–23. https://doi.org/10.3354/cr01309
- Haasnoot, M., Kwadijk, J., Van Alphen, J., Le Bars, D., Van Den Hurk, B., Diermanse, F., et al. (2020). Adaptation to uncertain sea-level rise; how uncertainty in Antarctic mass-loss impacts the coastal adaptation strategy of The Netherlands. *Environmental Research Letters*, 15(3), 034007. https://doi.org/10.1088/1748-9326/ab666c
- Haasnoot, M., Kwakkel, J. H., Walker, W. E., & ter Maat, J. (2013). Dynamic adaptive policy pathways: A method for crafting robust decisions for a deeply uncertain world. *Global Environmental Change*. https://doi.org/10.1016/j.gloenvcha.2012.12.006
- Haasnoot, M., van't Klooster, S., & van Alphen, J. (2018). Designing a monitoring system to detect signals to adapt to uncertain climate change. *Global Environmental Change*. https://doi.org/10.1016/j.gloenvcha.2018.08.003
- Haigh, I. D., Pickering, M. D., Green, J. A. M., Arbic, B. K., Arns, A., Dangendorf, S., et al. (2019). The tides they are a-changin': A comprehensive review of past and future non-astronomical changes in tides, their driving mechanisms and future implications. *Reviews of Geophysics*, 2018RG000636. https://doi.org/10.1029/2018RG000636
- Hallegatte, S. (2009). Strategies to adapt to an uncertain climate change. *Global Environmental Change*. https://doi.org/10.1016/j.gloenvcha.2008.12.003
- Hauer, M. E. (2017). Migration induced by sea-level rise could reshape the US population landscape. *Nature Climate Change*, 7(5), 321–325. https://doi.org/10.1038/nclimate3271
- Hermans, T. H. J., Gregory, J. M., Palmer, M. D., Ringer, M. A., Katsman, C. A., & Slangen, A. B. A. (2021). Projecting global mean sea-level change using CMIP6 models. *Geophysical Research Letters*, 48(5), e2020GL092064. https://doi.org/10.1029/2020GL092064
- Hermans, T. H. J., Katsman, C. A., Camargo, C. M. L., Garner, G., Kopp, R. E., & Slangen, A. B. A. (2022). The effect of wind stress on seasonal sea-Level change on the northwestern European shelf. *Journal of Climate*, 1–31. https://doi.org/10.1175/jcli-d-21-0636.1

#### Acknowledgments

The authors declare no competing financial interests. We thank all the researchers who have made their sea level data available: A. Garner for the sea-level database: and T. Wong, L. Jackson, R. Kopp, A. Nauels, D. Le Bars, A. Bakker, R. Kopp, S. Jevrejeva, J. Bamber, the SROCC Chapter 4 team for sharing their SLR time series and performing additional processing where needed. We thank Ilse van den Broek for figure design. This publication was supported by PROTECT. This project has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 869304, PROTECT contribution number 32

- Hermans, T. H. J., Tinker, J., Palmer, M. D., Katsman, C. A., Vermeersen, B. L. A., & Slangen, A. B. A. (2020). Improving sea-level projections on the Northwestern European shelf using dynamical downscaling. *Climate Dynamics*, 54(3–4). https://doi.org/10.1007/s00382-019-05104-5 Hinkel, J., Church, J. A., Gregory, J. M., Lambert, E., Le Cozannet, G., Lowe, J., et al. (2019). Meeting user needs for sea level rise information:
- A decision analysis perspective. *Earth's Future*, 7(3), 320–337. https://doi.org/10.1029/2018EF001071
   Horton, B. P., Khan, N. S., Cahill, N., Lee, J. S. H., Shaw, T. A., Garner, A. J., et al. (2020). Estimating global mean sea-level rise and its uncertainties by 2100 and 2300 from an expert survey. *Npj Climate and Atmospheric Science*, 3(1), 18. https://doi.org/10.1038/s41612-020-0121-5
- Horton, R., Little, C., Gornitz, V., Bader, D., & Oppenheimer, M. (2015). New York city panel on climate change 2015 report chapter 2: Sea level rise and coastal storms. *Annals of the New York Academy of Sciences*, *1336*(1), 36–44. https://doi.org/10.1111/nyas.12593
- Jackson, L. P., Grinsted, A., & Jevrejeva, S. (2018). 21st century sea-level rise in line with the Paris accord. *Earth's Future*. https://doi.org/10.1002/2017EF000688
- Jackson, L. P., & Jevrejeva, S. (2016). A probabilistic approach to 21st century regional sea-level projections using RCP and High-end scenarios. Global and Planetary Change, 146, 179–189. https://doi.org/10.1016/j.gloplacha.2016.10.006
- Jevrejeva, S., Grinsted, A., & Moore, J. C. (2014). Upper limit for sea level projections by 2100. *Environmental Research Letters*, 9(10), 104008. https://doi.org/10.1088/1748-9326/9/10/104008
- Jevrejeva, S., Jackson, L. P., Grinsted, A., Lincke, D., & Marzeion, B. (2018). Flood damage costs under the sea level rise with warming of 1.5°C and 2°C. Environmental Research Letters. https://doi.org/10.1088/1748-9326/aacc76
- Jevrejeva, S., Jackson, L. P., Riva, R. E. M., Grinsted, A., & Moore, J. C. (2016). Coastal sea level rise with warming above 2°C. Proceedings of the National Academy of Sciences, 13. https://doi.org/10.1073/pnas.1605312113
- Johansson, M. M., Pellikka, H., Kahma, K. K., & Ruosteenoja, K. (2014). Global sea level rise scenarios adapted to the Finnish coast. Journal of Marine Systems, 129, 35–46. https://doi.org/10.1016/j.jmarsys.2012.08.007
- Keeler, A. G., McNamara, D. E., & Irish, J. L. (2018). Responding to sea level rise: Does short-term risk reduction inhibit successful long-term adaptation? *Earth's Future*. https://doi.org/10.1002/2018EF000828
- Kirwan, M. L., Temmerman, S., Skeehan, E. E., Guntenspergen, G. R., & Fagherazzi, S. (2016). Overestimation of marsh vulnerability to sea level rise. *Nature Climate Change*. https://doi.org/10.1038/nclimate2909
- KNMI. (2021). KNMI Klimaatsignaal'21: hoe het klimaat in Nederland snel verandert (p. 72). KNMI.
- Knutti, R., Masson, D., & Gettelman, A. (2013). Climate model genealogy: Generation CMIP5 and how we got there. Geophysical Research Letters, 40, 1194–1199. https://doi.org/10.1002/grl.50256
- Kopp, R. E., DeConto, R. M., Bader, D. A., Hay, C. C., Horton, R. M., Kulp, S., et al. (2017). Evolving understanding of Antarctic ice-sheet physics and ambiguity in probabilistic Sea-Level projections. *Earth's Future*, 5(12), 1217–1233. https://doi.org/10.1002/2017EF000663
- Kopp, R. E., Horton, R. M., Little, C. M., Mitrovica, J. X., Oppenheimer, M., Rasmussen, D. J., et al. (2014). Probabilistic 21st and 22nd century sea-level projections at a global network of tide gauge sites. *Earth's Future*, 2, 383–406. https://doi.org/10.1002/2014EF000239
- Kopp, R. E., Kemp, A. C., Bittermann, K., Horton, B. P., Donnelly, J. P., Gehrels, W. R., et al. (2016). Temperature-driven global sea-level variability in the Common Era. Proceedings of the National Academy of Sciences, 113(11), E1434–E1441. https://doi.org/10.1073/pnas.1517056113
- Kwadijk, J. C. J., Haasnoot, M., Mulder, J. P. M., Hoogvliet, M. M. C., Jeuken, A. B. M., van der Krogt, R. A. A., et al. (2010). Using adaptation tipping points to prepare for climate change and sea level rise: A case study in The Netherlands. *Wiley Interdisciplinary Reviews: Climate Change*, https://doi.org/10.1002/wcc.64
- Lawrence, J., Bell, R., Blackett, P., Stephens, S., & Allan, S. (2018). National guidance for adapting to coastal hazards and sea-level rise: Anticipating change, when and how to change pathway. *Environmental Science & Policy*, 82, 100–107. https://doi.org/10.1016/j.envsci.2018.01.012
- Le Bars, D., Drijfhout, S., & De Vries, H. (2017). A high-end sea level rise probabilistic projection including rapid Antarctic ice sheet mass loss. *Environmental Research Letters*, 12, 044013. https://doi.org/10.1088/1748-9326/aa6512
- Le Cozannet, G., Nicholls, R. J., Hinkel, J., Sweet, W. V., McInnes, K. L., Van de Wal, R. S. W., et al. (2017). Sea level change and coastal climate services: The way forward. *Journal of Marine Science and Engineering*, 5(4). https://doi.org/10.3390/jmse5040049
- Masson, D., & Knutti, R. (2011). Climate model genealogy. Geophysical Research Letters, 38, L08703. https://doi.org/10.1029/2011GL046864
  Mcinnes, K. L., Church, J., Monselesan, D., Hunter, J. R., O'grady, J. G., Haigh, I. D., & Zhang, X. (2015). Information for Australian impact and adaptation planning in response to sea-level rise. Australian Meteorological and Oceanographic Journal, 65. https://doi.
- org/10.22499/2.6501.009
   Meinshausen, M., Raper, S. C. B., & Wigley, T. M. L. (2011). Emulating coupled atmosphere-ocean and carbon cycle models with a simpler
- model, MAGICC6 Part 1: Model description and calibration. Atmospheric Chemistry and Physics. https://doi.org/10.5194/acp-11-1417-2011 Mengel, M., Levermann, A., Frieler, K., Robinson, A., Marzeion, B., & Winkelmann, R. (2016). Future sea level rise constrained by observations
- and long-term commitment. *Proceedings of the National Academy of Sciences*, 113(10), 2597–2602. https://doi.org/10.1073/pnas.1500515113 Moss, R. H., Edmonds, J. A., Hibbard, K. A., Manning, M. R., Rose, S. K., van Vuuren, D. P., et al. (2010). The next generation of scenarios for
- climate change research and assessment. *Nature*, 463(747-756), 747-756. https://doi.org/10.1038/nature08823 Nauels, A., Meinshausen, M., Mengel, M., Lorbacher, K., & Wigley, T. M. L. (2017). Synthesizing long-term sea level rise projections – the
- MaGICC sea level model v2.0. Geoscientific Model Development, 10(6), 2495–2524. https://doi.org/10.5194/gmd-10-2495-2017
  Nauels, A., Rogelj, J., Schleussner, C.-F., Meinshausen, M., & Mengel, M. (2017). Linking sea level rise and socioeconomic indicators under the
- sauels, A., Kogelj, J., Schleussner, C.-F., Meinshausen, M., & Mengel, M. (2017). Linking sea level rise and socioeconomic indicators under the Shared Socioeconomic Pathways. *Environmental Research Letters*, 12(11), 114002. https://doi.org/10.1088/1748-9326/aa92b6
- Nepf, H. M. (2012). Hydrodynamics of vegetated channels. Journal of Hydraulic Research, 50(3), 262–279. https://doi.org/10.1080/00221686. 2012.696559
- O'Neill, B. C., Kriegler, E., Ebi, K. L., Kemp-Benedict, E., Riahi, K., Rothman, D. S., et al. (2017). The roads ahead: Narratives for shared socioeconomic pathways describing world futures in the 21st century. *Global Environmental Change*, 42, 169–180.
- Oppenheimer, M., Glavovic, B. C., Hinkel, J., van de Wal, R., Magnan, A. K., Abd-Elgawad, A., et al. (2019). Chapter 4: Sea level rise and implications for low lying islands, coasts and communities. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska (Eds.), *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate*.
- Palmer, M. D., Gregory, J. M., Bagge, M., Calvert, D., Hagedoorn, J. M. M., Howard, T., et al. (2020). Exploring the drivers of global and regional sea-level change over the 21st century and beyond. *Earth's Future*, 8, e2019EF001413. https://doi.org/10.1029/2019EF001413
- Palmer, M. D., Howard, T., Tinker, J., Lowe, J. A., Bricheno, L., Calvert, D., et al. (2018). UKCP18 marine projection report [UK climate projections (UKCP)]. Met Office.
- Pattyn, F., & Morlighem, M. (2020). The uncertain future of the Antarctic Ice Sheet. Science. https://doi.org/10.1126/science.aaz5487
- Perrette, M., Landerer, F., Riva, R., Frieler, K., & Meinshausen, M. (2013). A scaling approach to project regional sea level rise and its uncertainties. *Earth System Dynamics*, *4*, 11–29. https://doi.org/10.5194/esd-4-11-2013



- Rasmussen, D. J., Bittermann, K., Buchanan, M. K., Kulp, S., Strauss, B. H., Kopp, R. E., & Oppenheimer, M. (2018). Extreme sea level implications of 1.5°c, 2.0°c, and 2.5°c temperature stabilization targets in the 21st and 22nd centuries. *Environmental Research Letters*, 13(3). https:// doi.org/10.1088/1748-9326/aaac87
- Roberts, C. D., Calvert, D., Dunstone, N., Hermanson, L., Palmer, M. D., Smith, D., et al. (2016). On the drivers and predictability of seasonal-to-interannual variations in regional sea level. *Journal of Climate*. https://doi.org/10.1175/JCLI-D-15-0886.1
- Schleussner, C. F., Lissner, T. K., Fischer, E. M., Wohland, J., Perrette, M., Golly, A., et al. (2016). Differential climate impacts for policy-relevant limits to global warming: The case of 1.5°c and 2°c. *Earth System Dynamics*. https://doi.org/10.5194/esd-7-327-2016
- Simpson, M. J. R., Nilsen, J. E. Ø., Ravndal, O. R., Breili, K., Sande, H., Kierulf, H. P., et al. (2015). Sea level change for Norway. Past and present observations and projections to 2100.
- Slangen, A. B. A., Adloff, F., Jevrejeva, S., Leclercq, P. W., Marzeion, B., Wada, Y., & Winkelmann, R. (2017). A review of recent updates of sea-level projections at global and regional scales. *Surveys in Geophysics*, 38(1), 385–406. https://doi.org/10.1007/s10712-016-9374-2
- Slangen, A. B. A., Carson, M., Katsman, C. A., van de Wal, R. S. W., Köhl, A., Vermeersen, L. L. A., & Stammer, D. (2014). Projecting twenty-first century regional sea-level changes. *Climatic Change*, 124(1–2), 317–332. https://doi.org/10.1007/s10584-014-1080-9
- Slangen, A. B. A., Katsman, C. A., van de Wal, R. S. W., Vermeersen, L. L. A., & Riva, R. E. M. (2012). Towards regional projections of twenty-first century sea-level change based on IPCC SRES scenarios. *Climate Dynamics*, 38(5–6), 1191–1209. https://doi.org/10.1007/ s00382-011-1057-6
- Slangen, A. B. A., Meyssignac, B., Agosta, C., Champollion, N., Church, J. A., Fettweis, X., et al. (2017). Evaluating model simulations of twentieth-century sea level rise. Part I: Global mean sea level change. *Journal of Climate*, 30(21), 8539–8564. https://doi.org/10.1175/ JCLI-D-17-0110.1
- Sriver, R. L., Lempert, R. J., Wikman-Svahn, P., & Keller, K. (2018). Characterizing uncertain sea-level rise projections to support investment decisions. PLoS One, 13(2). https://doi.org/10.1371/journal.pone.0190641
- Stephens, S. A., Bell, R. G., & Lawrence, J. (2018). Developing signals to trigger adaptation to sea-level rise. *Environmental Research Letters*, 13(10), 104004. https://doi.org/10.1088/1748-9326/aadf96
- Sweet, W. V., Kopp, R. E., Weaver, C. P., Obeysekera, J., Horton, R., Thieler, E. R., & Zervas, C. (2017). Global and regional sea level rise scenarios for the United States.
- Wahl, T., Haigh, I. D., Nicholls, R. J., Arns, A., Dangendorf, S., Hinkel, J., & Slangen, A. B. A. (2017). Understanding extreme sea levels for broad-scale coastal impact and adaptation analysis. *Nature Communications*, 8. https://doi.org/10.1038/ncomms16075
- Wong, T. E., Bakker, A., Ruckert, K., Applegate, P., Slangen, A., & Keller, K. (2017). BRICK v0.1, a simple, accessible, and transparent model framework for climate and regional sea-level projections. *Geoscientific Model Development Discussions*. https://doi.org/10.5194/ gmd-2016-303